

Comments:

This letter evaluates the article by Gravina et al. on ChatGPT's potential in providing medical information for inflammatory bowel disease patients, and calls for more nuanced evaluation methods to truly gauge large language models' capabilities in clinical applications.

Please find enclosed the resubmitted revised manuscript

We have read the reviewer comments. However, there is no requirement for the amendment of manuscript.

I am herewith attaching the revised manuscript.

I always do appreciate very much your support.

With my best regards,

Sincerely,